Author's response to reviews

Title: Ventricular Assist Devices and Non-Cardiac Surgery

Authors:

S. Michael Roberts DO (sroberts4@hmc.psu.edu)  
David G Hovord BA MB BChir (davidhov@med.umich.edu)  
Ramesh Kodavatiganti MD (kodavatigantir@email.chop.edu)  
Subramanian Sathishkumar MBBS (ssathish@med.umich.edu)

Version: 5  
Date: 31 August 2015

Author's response to reviews:

Following changes as requested were made:

1. We renamed introduction as Background. (line 49)
2. Manuscript restructured according to author’s instructions.
3. Authors contribution section revised. (line 433-437)
4. Figure legends for all four figures added to the end of the manuscript.
5. Files 1 and 2 removed as requested.
6. Confirmation of permissions for figures used added to manuscript.

Dr Subramanian Sathishkumar